Author's response to reviews

Title: Antinociceptive activity of methanolic extract of Muntingia calabura leaves: Further elucidation of the possible mechanisms

Authors:

Zainul Amiruddin Zakaria (dr_zaz@yahoo.com)
Mohd. Hijaz Mohd. Sani (hijaz.mohd@gmail.com)
Muhammad Nazrul Hakim Abdullah (nazrul-hakim@gmail.com)
Zuraini Ahmad (zuraini@upm.edu.my)
Arifah Abdul Kader (arifah@upm.edu.my)
Lay Kek Teh (tehlaykek@yahoo.com)
Mohd. Zaki Salleh (zakisalleh.mzs@gmail.com)

Version: 2 Date: 17 October 2013

Author's response to reviews:

The ethics approval has been incorporated into the manuscript.
The figures legend have been transferred into the manuscript.